CSE 322: In tro duction to F ormal Mo dels in Computer Science

Assignmen t #7

Ma y 27, 1998

due: W ednesda y , June 3

1. Exercise 3.4.1

2. Exercise 3.5.8. Use the v ersion of the Pumping Lemma pro v ed in class,

rather than Exercis e 3.5.7.

3. Exercise 3.5.14 (a), (b), and (d). Hin ts: Use closure prop erties lib erally .

Do part (d) b efore part (b).

4. Exercise 3.5.15

5. Exercise 3.6.1. Do this problem using the handouts rather than the

textb o ok v ersi ons of Chomsky normal form and the dynamic program-

ming algorithm. Instead of the string x giv en in the exercise , use the

string

id + (id + id) � id

Also sho w ho w to reco v er the parse tree for this string from y our dy-

namic programming table.